Author's response to reviews

Title: The effectiveness of motorized lumbar traction in the management of LBP with lumbo sacral nerve root involvement: A feasibility study.

Authors:

Annette A Harte (aa.harte@ulster.ac.uk)
David G Baxter (physio.dean@otago.ac.uk)
Jacqueline H Gracey (jh.gracey@ulster.ac.uk)

Version: 4 Date: 13 November 2007

Author's response to reviews:

Thank you for accepting this manuscript.
I have now dealt with the requests made by Helen Frost as follows

1. Conclusion of abstract the words ¿high quality¿ removed.
2. Consort diagram changed to numbers completing trial as opposed to numbers analysed.
3. In section on Statistical analysis I have reworded the information on ITT and per protocol analysis and hopefully this is now clearer
4. Pg 12 alteration made in wording as requested.
5. I have checked and corrected some minor typing errors.

I have also checked formatting as requested by you and added a title page as requested. I hope I have met all the revisions but please don¿t hesitate to contact me if there are any omissions. Once again thank you for your time and patience with this article and for the helpful reviews which has enabled this paper to be structured accordingly.